Author's response to reviews

Title: A longitudinal study investigating quality of life and nutritional outcomes in advanced cancer patients receiving home parenteral nutrition

Authors:

Pankaj Vashi (pankaj.vashi@ctca-hope.com)
Sadie Dahlk (sadie.dahlk@ctca-hope.com)
Brenten Popiel (brenten.popiel@ctca-hope.com)
Carolyn Lammersfeld (carolyn.lammersfeld@ctca-hope.com)
Carol Iretion-Jones (dr.cijrd@verizon.net)
Digant Gupta (digant.gupta@ctca-hope.com)

Version: 2  Date: 14 March 2014

Author's response to reviews:

Line numbering has been provided as requested.